Author's response to reviews

Title: Right axillary and femoral artery perfusion with mild hypothermia for aortic arch replacement

Authors:

Jige Guo (jigeqxuo@yeah.net)
Yue Wang (yuewangc@yeah.net)
Jihong Zhu (jiecso@yeah.net)
Jie Cao (jihongzg@yeah.net)
Zili Chen (zilichnn@yeah.net)
Zhijun Li (zhijuli@yeah.net)
Ximing Qian (jgdoccn@126.com)

Version: 4 Date: 14 March 2014

Author's response to reviews: see over
Dear Editor:

We would like to submit this manuscript the manuscript entitled “Right axillary and femoral artery perfusion with mild hypothermia for aortic arch replacement”. We would like to submit this manuscript as a scientific contribution for publication in Journal of Cardiothoracic Surgery. This result is very interesting and may be of particular interest to the readers of your journal. This manuscript is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal.

The authors declare that they have no conflict of interests. Its publication has been approved by all co-authors.

If there are any questions and omissions with this submission, please let me know.

Thanks.

Ximing Qian,

Department of Cardiothoracic Surgery,

Sir Run Run Shaw Hospital, College of Medicine, Zhejiang University,

No. 3 Qingchundong Road,

Hangzhou 310016

China

Tel: +86 571 86006158

Fax: +86 571 86006158

E-mail: jgdoccn@126.com